Author’s response to reviews

Title: Qualitative study exploring healthy eating practices and physical activity among adolescent girls in rural South Africa

Authors:

Heather M Sedibe (sedibemh@gmail.com)
Kathleen Kahn (Kathleen.Kahn@wits.ac.za)
Kerstin Edin (kerstin.edin@umu.se)
Tabby Gitau (tabbymuthoni@gmail.com)
Anneli Ivarsson (anneli.ivarsson@epiph.umu.se)
Shane A Norris (Shane.Norris@wits.ac.za)

Version: 8 Date: 18 August 2014

Author’s response to reviews: see over
c/o Gilbert Tacbobo
Journal Editorial Office
BioMed Central

Dr. Kathryn Parkinson and Ms. Catherine Olino

Re: Resubmission of manuscript MS: 3341193231208254: Qualitative study exploring healthy eating practices and physical activity among adolescent girls in rural South Africa.

Thank you for allowing us to re-submit a revision of the above mentioned manuscript and the comments from the reviewers which we have helped improve our manuscript. Please accept the attached revised manuscripts where comments raised by reviewers have been addressed.

Yours sincerely

Heather Sedibe (Mrs)
modiehi.sedibe@ul.ac.za
Cell: 27- 82 507 7719

Student: University of Witswatersrand
Faculty of Health Sciences, Department of Paediatrics,
MRC/WITS Developmental Pathways for Health Research Unit,
Johannesburg, South Africa
Comments and feedback:

1. Author emails
   Please ensure that each author is represented by a single email and that this is correct and identical on both the authors details page of the submission system and on the title page of the manuscript. We note that author Sedibe has two different email addresses. – the email address of  Ms Sedibe has been corrected and updated on the submission system.

2. Abstract
   Please ensure that your abstracts are identical on the 'Manuscript Details' page of the submission system and in the main body of the manuscript. These are currently different. Please see the following link for formatting your abstract:
   http://www.biomedcentral.com/bmcpediatr/authors/instructions/researcharticle#formatting-abstract – The correct abstract has been loaded on the “manuscript details” page

3. RATS
   Please include a statement in your manuscript confirming that your research has adhered to the guidelines for Qualitative research review guidelines (RATS) as outlined here:
   http://www.biomedcentral.com/authors/rats - The statement has been added on page 7 line 6